

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Kuhn, Andreas

Working Paper

Redistributive preferences, redistribution, and inequality: Evidence from a panel of OECD countries

IZA Discussion Papers, No. 6721

Provided in Cooperation with:

IZA - Institute of Labor Economics

Suggested Citation: Kuhn, Andreas (2012): Redistributive preferences, redistribution, and inequality: Evidence from a panel of OECD countries, IZA Discussion Papers, No. 6721, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/62555

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





IZA DP No. 6721

Redistributive Preferences, Redistribution, and Inequality: Evidence from a Panel of OECD Countries

Andreas Kuhn

July 2012

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Redistributive Preferences, Redistribution, and Inequality: Evidence from a Panel of OECD Countries

Andreas Kuhn

University of Zurich and IZA

Discussion Paper No. 6721 July 2012

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit organization supported by Deutsche Post Foundation. The center is associated with the University of Bonn and offers a stimulating research environment through its international network, workshops and conferences, data service, project support, research visits and doctoral program. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ABSTRACT

Redistributive Preferences, Redistribution, and Inequality: Evidence from a Panel of OECD Countries*

This paper describes individuals' inequality perceptions, distributional norms, and redistributive preferences in a panel of OECD countries, primarily focusing on the association between these subjective measures and the effective level of inequality and redistribution. Not surprisingly, the effective level of redistribution (after tax-and-transfer inequality) is positively (negatively) correlated with redistributive preferences. There is also evidence showing that the subjective and objective dimension of inequality and redistribution are, at least partially, linked with individuals' political preferences and their voting behavior. The association between objective and subjective measures of inequality and redistribution vanishes, however, once more fundamental country characteristics are taken into account. This suggests that these characteristics explain both redistributive preferences as well as the effective level of redistribution and after tax-and-transfer inequality.

JEL Classification: D31, D63, J31

Keywords: inequality perceptions, distributional norms, redistributive preferences,

inequality, redistribution, political preferences

Corresponding author:

Andreas Kuhn
University of Zurich
Department of Economics
Mühlebachstrasse 86
8008 Zurich
Switzerland

E-mail: andreas.kuhn@econ.uzh.ch

_

I thank Rafael Lalive, Andrew Oswald, and Josef Zweimüller for helpful discussions at an early stage of this project. I also thank Simon Büchi, Sandro Favre, and Andreas Steinhauer for superb research assistance. Financial support by the Austrian Science Fund (FWF) is gratefully acknowledged (S 10304-G16: "The Austrian Center for Labor Economics and the Analysis of the Welfare State").

1 Introduction

The redistribution of income and wealth is an issue of utmost economic and political significance, as a substantial amount of resources is redistributed in all OECD member countries. In the mid-2000s, for example, overall public cash benefits amounted to 15.8%, and household taxes to 31.1% of average household income among working age people (OECD, 2008). Even more remarkably, public cash benefits (as a percentage of household income) among working age individuals range from a low of about 6% in the United States to a high of as much as 30% in Poland, while household taxes range from a low of about 21% in Ireland to a high of about 54% in Denmark and Iceland. Looking at these numbers, it is both important and natural to ask why countries differ so much with respect to the amount of resources that are redistributed and, consequently, the level of after tax-and-transfer inequality.¹

It seems very natural to take country differences in redistributive preferences as a starting point when trying to explain the variation in the effective level of redistribution across countries. Indeed, many researchers appear to share the view that individuals' subjective perceptions and attitudes play an important role in explaining the observed country differences in inequality and redistribution. For example, Alesina and Angeletos (2005, p.960) argue that "(...) the difference in political support for redistribution appears (...) to reflect a difference in social perceptions regarding the fairness of market outcomes and the underlying sources of income inequality". Such arguments are backed by empirical evidence from Alesina et al. (2001), for example, who show that there is a close relationship between social expenditure and the belief that luck is important in determining one's income at the aggregate level. Consistent with this evidence, some recent theoretical papers have forcefully pushed the idea that individuals' redistributive preferences and the effective amount of redistribution should be viewed as simultaneously determined and thus in equilibrium, at least in the short-run. In Piketty (1995), individuals imperfectly learn about the relative importance of effort and luck in determining their own income from their own experience and, depending on their beliefs, decide on their preferred tax rate. The optimal level of effort will in turn depend on the chosen tax rate. The

¹See Doerrenberg and Peichl (2012) for an econometric analysis of the association between explicit redistributive policies (e.g. public social expenditure) and the reduction in income inequality effectively attained.

setup is similar in the model of Alesina and Angeletos (2005), where different beliefs about the relative role of ascribed versus acquired characteristics in determining individual earnings lead voters to choose different tax rates. The tax rate, in turn, will determine the monetary return on individual effort. The effective return on individual effort will then either reinforce or weaken the belief in the relative importance of individual effort. The basic setup is similar in the model of Bénabou and Tirole (2006), where individuals feel a need to believe in a just world because such a belief helps them convince themselves, and/or their children, that personal effort will pay off. If a majority of individuals ends up with such beliefs, they will implement a low tax rate. At the same time, if individuals expect taxes to be low, this will strengthen their belief that individual effort will ultimately materialize. One important common feature of these models is that multiple equilibria are possible, and thus they are all able to reproduce the stylized distinction between Europeans and Americans, i.e. a "European" equilibrium where the belief is prevalent that luck determines income, with a high demand for redistribution, and consequently a high level of redistribution; as well as an "American" equilibrium where the belief in one's effort is prevalent, with a low level of redistributive preferences, and with a correspondingly low amount of redistribution.

Somewhat surprisingly, however, empirical evidence on the association between redistributive preferences and the effective level of redistribution is quite limited.² One of the available studies using cross-country data is Corneo and Grüner (2002). Even they do not mainly focus on differences across countries, they do present some evidence suggesting that there are large differences in redistributive preferences across different countries. Specifically, they find large differences in redistributive preferences between former socialist countries and Western democracies. Probably the most comprehensive descriptive evidence on country differences in redistributive preferences to date comes from Osberg and Smeeding (2006). Interestingly, they show that there are many more features in the variation of redistributive preferences across countries than the contrast between the US and Europe. For example, they show that there is more polarization of attitudes and less concern for reducing wage differentials at the bottom

²Most of the available empirical studies focus on the determinants of individual-level preferences for redistribution, most often only using data from one single country (e.g. Alesina and Giuliano, 2009; Alesina and La Ferrara, 2005; Fong, 2001; Ravallion and Lokshin, 2000).

of the wage distribution among Americans than among Europeans. While they use a similar conceptual framework to that which I use in this paper, they do not relate country differences in their measure of redistributive preferences to differences in objective measures of inequality or redistribution. Isaksson and Lindskog (2009) focus on cross-country variation in the effect of social norms on preferences for redistribution. They find that the association between redistributive preferences and social norms differs widely across countries, and that these differences explain part of the observed country differences in redistributive preferences. Again, however, they do not relate their subjective measure of redistributive preferences with some objective measure of either the effective level of inequality or the effective level of redistribution.

This paper builds on and extends the existing empirical literature using internationally comparable survey data covering eighteen different OECD member countries (regions within the same country in a few cases, such as former East and West Germany) and three different points in time (i.e. 1987, 1992, and 1999). Moreover, the empirical analysis is based on a simple yet intuitive conceptual framework which is almost ideally suited for tackling the following issues, which also present the main contributions of this paper. First, it presents extensive empirical evidence for this significant group of countries on the hypothesized association of individuals' subjective perceptions of inequality and distributional norms with the effective level of inequality and the effective extent of redistribution. At the aggregate level, I find that the difference in the Gini coefficient of disposable household income before and after taxes and transfers, total public social expenditure, as well as after tax-and-transfer household income inequality are strongly and significantly associated with individuals' redistributive preferences. Second, this paper provides evidence on how these two dimensions might be linked together at the individual level. Specifically, I show that there is a strong association between individuals' subjective evaluations of inequality and their more general political preferences, such as their support for progressive taxation. Third and finally, my empirical analysis also shows that this association vanishes once additional, and arguably more fundamental country characteristics, such as specific political institutions or ethnic fractionalization, are taken into account. Taken together, these results suggest that while redistributive preferences and the effective level of redistribution are in a short-run equilibrium, they are both ultimately driven by more fundamental national

characteristics such as a country's political institutions.

The remainder of this paper is organized as follows. The next section presents the main data source. Section 3 briefly describes the conceptual framework. Section 4 presents descriptive evidence on the diversity of inequality perceptions and distributional norms both across countries and over time. Section 5 provides econometric evidence on the hypothesized association between subjective and objective inequality measures, as well as on the association between subjective inequality measures and individuals' more general political preferences, representing one potential mechanism linking the two dimensions. Section 6 concludes.

2 Data

I primarily rely on data from three surveys on the causes and consequences of social inequality administered by and available from the International Social Survey Program (ISSP).³ The first survey on social inequality was run in 1987, the second in 1992, and the third followed in 1999. Moreover, a fourth survey was administered in 2009, but these data are not yet (but should soon be) available to researchers. The number of participating countries has steadily increased over time. While only ten countries participated in the first survey, the number of participating countries has increased steadily and about thirty countries participated in the 1999 survey. Because of restricted data availability for some additional aggregate-level variables, I had to restrict the analysis to 21 OECD member countries (or, in a few cases, regions within the same country, such as East and West Germany). Moreover, because the number of participating countries has steadily increased over time, the data are unbalanced on the longitudinal dimension and I end up with a maximum of 38 observations in the aggregate level analysis. Table 1 shows the number of available observations, both at the individual and the aggregate levels.

Table 1

³The ISSP is a collaboration of various national survey organizations which administers one survey a year with an alternating core theme such as religion, work orientations, or social inequality. See the organization's homepage for more information: www.issp.org.

I will use some additional, aggregate-level variables from various other sources, such as aggregate income or the effective level of inequality (measured as the Gini coefficient in disposable household income, either before or after taxes and transfer payments) in the empirical analysis. These additional variables, and their sources, are listed in appendix table A.1, along with some basic descriptives.

3 Conceptual Framework

In will now discuss how to construct subjective measures of inequality perceptions and distributional norms, as well as a measure of individuals' normative assessment of market justice, drawing on a simple conceptual framework proposed by and discussed in more detail in Kuhn (2011). The framework is fundamentally based on individuals' subjective wage estimates for people working in different occupations, such as a bus driver, a skilled worker in a factory, or a doctor in general practice (see appendix B for the exact wording of the questions and additional details).

3.1 Objective Wages

The natural starting point is the conventional measurement of objective wage inequality, however. One of the most routinely used and best known inequality measures is the Gini coefficient. The computation of the Gini coefficient is usually based on individual-level wage data, but it is possible to approximate the individual-level Gini coefficient using group-level data on wages (e.g. Gastwirth and Glauberman, 1976). Indeed, observing wage information for only two distinct groups of individuals suffices in principle for approximating the underlying inequality of individual wages.

Formally, assume that there are only two distinct, mutually exclusive, and exhaustive groups of wage earners (labeled bottom and top group, respectively, below) and that we observe the following information describing the distribution of wages across these two groups:

$$y \equiv \left(\overline{y}_{\text{bottom}}, \overline{y}_{\text{top}}, f_{\text{bottom}}\right),\tag{1}$$

with $\overline{y}_{\text{bottom}}$ and $\overline{y}_{\text{top}}$ denoting the average wage within the bottom and the top group of wage earners, respectively, and with f_{bottom} denoting the fraction of the population that belongs to the bottom group.⁴ The Gini coefficient in this simplified setup turns out to be equal to the difference between the population share of the bottom group and the wage share of the bottom group (see Kuhn, 2011):⁵

$$G = f_{\text{bottom}} - q_{\text{bottom}} \tag{2}$$

In the scenario with only two distinct groups of wage earners, the Gini coefficient is thus simply given by the difference between the population share and the wage share of the bottom group.

3.2 Subjective Wage Estimates

The simplified setup formalized by equations (1) and (2) can also be applied to the case of subjective wage estimates which are available from the ISSP data, subject to two modifications. The first modification is due to the fact that there are two conceptually distinct wage estimates from an individual's subjective point of view: actual and ethical wage estimates. While actual wage estimates refer to wages that people perceive to actually prevail within a given group, ethical wage estimates refer to wages they would judge as legitimate the same given group. The second modification is simply due to the fact that both actual and ethical wage estimates will usually differ across individuals. This implies that the wage shares of the two groups, and thus subjective inequality measures in general, become individual-specific quantities. We

⁴Because there are only two groups of wage earners, and because these two groups represent the overall population of wage earners, the two population shares must add up to one. This implies that the two population shares are given by f_{bottom} and $(1 - f_{\text{bottom}}) = f_{\text{top}}$, respectively.

⁵Note that the average wage in the population is given by $\overline{y} = \overline{y}_{\text{bottom}} \cdot f_{\text{bottom}} + \overline{y}_{\text{top}} \cdot f_{\text{top}}$. This implies that the wage share of the bottom group is given by $q_{\text{bottom}} = (f_{\text{bottom}} \cdot \overline{y}_{\text{bottom}})/\overline{y}$, and thus that the Gini coefficient can be computed based on the information contained in vector y alone. Moreover, it is always the case that $\overline{y}_{\text{bottom}} \leq \overline{y}_{\text{top}}$ for objective wage data. Thus q_{bottom} is always smaller than or equal to f_{bottom} and, consequently, G always lies between 0 and f_{bottom} .

⁶The conceptual distinction between individuals' inequality perceptions and their distributional norms is prevalent in the sociological literature; see Jasso (1980) or Osberg and Smeeding (2006) among many others.

 $^{^{7}}$ Even though one can imagine that individuals also have different perceptions and/or norms about the population shares of the two groups, I will treat them as fixed parameters, which simply implies that f_{bottom} does not vary across individuals. The main reason for doing so is that there is no adequate information in the survey which could be used to plausibly approximate individuals' perceptions of population shares. Moreover, treating f_{bottom} as a fixed parameter also makes it possible to exclusively focus on differences in wage estimates

thus end up with two distinct distributions of inequality indices in the case of subjective wage estimates, while it is possible to measure objective wage inequality with one single number only.

Formally, assume that the following two triplets of information are observed for each respondent – perfectly analogous to the case of objective wage information, besides the two modifications discussed above (compare with equation (1)):

$$y(i)^{\text{actual}} \equiv (\overline{y}(i)^{\text{actual}}_{\text{bottom}}, \overline{y}(i)^{\text{actual}}_{\text{top}}, f_{\text{bottom}}), \text{ and }$$
 (3a)

$$y(i)^{\text{ethical}} \equiv (\overline{y}(i)^{\text{ethical}}_{\text{bottom}}, \overline{y}(i)^{\text{ethical}}_{\text{top}}, f_{\text{bottom}}),$$
 (3b)

with $y(i)^{\text{actual}}$ denoting the set of information that describes an individual's perception of the actual wage distribution and with $y(i)^{\text{ethical}}$ referring to those wages that she would judge as fair.⁸ Based on (3a) and (3b), respectively, the corresponding individual-level Gini coefficients can be easily computed as follows:

$$G(i)^{\text{actual}} = f_{\text{bottom}} - q(i)^{\text{actual}}_{\text{bottom}}, \quad \text{and}$$
 (4a)

$$G(i)^{\text{ethical}} = f_{\text{bottom}} - q(i)^{\text{ethical}}_{\text{bottom}},$$
 (4b)

where $q(i)_{\text{bottom}}^{\text{actual}}$ denotes the actual share of the total wage bill going to the bottom group and $q(i)_{\text{bottom}}^{\text{ethical}}$ denotes the wage share that the bottom group judges as fair. Thus, as in the case of objective wage data, the Gini coefficient is simply given by the difference between the population share of the bottom group and the wage share of the bottom group.⁹ $G(i)^{\text{actual}}$ and $G(i)^{\text{ethical}}$ represent, respectively, individual perceptions of inequality in market wages and distributional norms with respect to market wages (inequality perceptions and distributional norms, for short).

since, if f_{bottom} is treated as a fixed parameter across individuals, all variation in the two subjective inequality measures must be due to variation in subjective wage estimates.

⁸Appendix B details how the different components of equations (3a) and (3b) can be estimated from individual-level wage estimates for people working in different occupations that are available in the ISSP data.

⁹In principle, the two subjective Gini coefficients can take on negative values (in contrast to the Gini coefficient describing objective wage data) because some individuals may think that the wage share of the bottom group is actually larger than their population share (i.e. $q(i)_{\text{bottom}}^w$ can take on any value between zero and one). As shown in table 2 below, this is indeed true for a tiny fraction of the sample (0.1% in the case of actual and 0.5% in the case of ethical wage estimates).

If both inequality perceptions and distributional norms are observed for the same individual, it is also possible to define an individual's redistributive preferences as her desired relative reduction in the perceived level of inequality in market wages:

$$R(i) = \left(1 - \frac{G(i)^{\text{ethical}}}{G(i)^{\text{actual}}}\right),\tag{5}$$

with $G(i)^{\text{actual}}$ and $G(i)^{\text{ethical}}$ as defined in equation (4a) and (4b), respectively. Note that a positive demand for some equalization of market wages can only arise if the evaluation of ethical wages differs from the perceived distribution of market wages. Finally, note that R(i) measures only the potential demand for redistribution because the measure does not directly imply that individuals actually desire that the distribution of market wages be adjusted according to their evaluations. Thus, while R(i) is directly informative about the discrepancy between individuals' perceptions of actual wage inequality and their normative views of the fair distribution of wages, it is not necessarily informative regarding their beliefs that something should be done to eliminate this discrepancy, or even more specifically that the state should intervene correspondingly. It may thus be more appropriate to think of R(i) as a measure of individuals' normative assessment of market justice, or rather the absence of market justice, with values of R(i) close to (far from) zero indicating a strong (weak) belief in market justice.

4 The Diversity of Inequality Perceptions, Distributional Norms, and Redistributive Preferences

4.1 Main Distributional Features

I first present some simple descriptive statistics related to individuals' inequality perceptions and their distributional norms in the data pooled across all countries and years, fleshing out the most important distributional features of the two subjective inequality indices and redistributive preferences.

¹⁰Note that the demand for equalization of market wages can take on negative values, namely if the ethical inequality index is larger the perceived inequality index. The demand for equalization of market wages can also be larger than one if either $G(i)^{\text{actual}}$ or $G(i)^{\text{ethical}}$ takes on a negative value (cf. footnote 9).

Table 2

These statistics are shown in table 2. Panel (a) first shows the estimated fraction of individuals (in any given country and survey year) who are classified as belonging to the bottom group of wage earners, f_{bottom} . The estimated population share of the bottom group equals 77% on average, varying between a low of about 57% (Canada, 1992) to a high of almost 92% (Poland, 1992). The average actual wage share of the bottom (top) group amounts to 42% (58%), while the average ethical wage share is about 54% (about 46%).¹¹

Panel (b) shows descriptives for the two subjective inequality indices. Average inequality perception equals 0.451 across all countries/regions and years. Also note that there is not one single individual who does not perceive any wage differentials at all (this is indicated by the fact that the fraction of individuals who perceive the wages of the bottom group to be the same as the wages of the top group is zero). At the same time, average ethical inequality only amounts to 0.301, i.e. the ethical level of inequality is about a third lower than the perceived level. Moreover, note that only few individuals (less than 1%) judge absolute equality as fair, as can be seen from the fraction of individuals with an ethical inequality of zero.¹²

Panel (c) shows that people favor a more equal distribution of market wages across occupations than the distribution they perceive to actually exist (on average by about one-third). Indeed, the overwhelming majority (roughly 90%) of individuals has a positive demand for some equalization of wages, while only a small fraction of the sample has no or even a negative demand for equalizing market wages (about 8.2% and 1.6%, respectively). At the same time, table 2 also shows that only a few individuals (less than 1%) would like to eliminate all differences in wages across different occupations (i.e. have a demand for equalization of market

¹¹One interesting feature not evident in table 2 is the fact that the overall ethical wage is on average lower than the overall actual wage in most countries and years.

¹²It seems worth mentioning that the overall averages of the two subjective Gini coefficients are surprisingly close to mean levels of inequality before and after taxes and transfers, respectively (see appendix table A.1). While I certainly do not want to stress this feature too much, not the least because subjective measures relate to individual wages while the objective measures relate to household incomes, the similarity between mean inequality perceptions and inequality before taxes and transfers does suggest that individuals appear to have, on average, relatively realistic perceptions of wages for people working in different occupations. See also Osberg and Smeeding (2006) on this issue.

¹³A negative demand for equalizing market wages is suggestive of a desired regressive transfer. In most cases, however, a negative value simply results from an individual's desire to increase both the wage of the bottom and the top group, but with a larger desired relative increase for the top group (note that overall perceived wages may be different from overall ethical wages; cf. footnote 11).

wages exactly equal to one).

4.2 Differences Across Regions and Countries

I will next look at differences in these measures across broadly defined regions, as well as over countries within these regions, and across time. Indeed, one of the most recurrent themes in the existing literature is the pronounced difference between Europe and the US regarding redistributive preferences (e.g. Alesina and Glaeser, 2004; Alesina et al., 2001). I thus start with a comparison between the Anglo-American and the European countries (while separating Eastern and Western European countries, however).

Figure 1

The upper three panels of figure 1 show density estimates of inequality perceptions, distributional norms, and redistributive preferences for each the three regions. The densities differ markedly, and in the expected direction, from each other for each of the three measures. First, as shown in panel (a), individuals in the Anglo-American countries perceive much higher wage inequality than both Eastern and Western Europeans. The figure also shows, however, that the distribution of inequality perceptions among Eastern Europeans is very different from those of Western Europeans. Panel (b) shows that Europeans are much less tolerant towards inequality than individuals from Anglo-American countries, but also that the tolerated wage inequality is lowest among Eastern Europeans. Regarding redistributive preferences, panel (c) shows that individuals from the Anglo-American countries indeed have the lowest demand for equalization of market wages, and that Western Europeans tend to have less demand for equalization of market wages than Eastern Europeans.

However, as Osberg and Smeeding (2006) and others point out, there is considerable heterogeneity among European and Anglo-American countries as well. The remaining three panels of figure 1 illustrate the heterogeneity in perceptions and beliefs across countries within the three broader regions. Each panel shows country-specific density estimates of the demand for equalization of market wages (similar regional variation is found for the two subjective inequality measures; results not shown, however). Only two countries within each region are specifically

marked for better readability. Panel (d) shows density estimates among the group of Western European countries. In this region, Sweden and Switzerland represent the two extreme distributions, while the remaining countries all fall somewhere between these two countries. Correspondingly, East Germany and Hungary represent the two extremes among the Eastern European countries, shown in panel (e), and so do Northern Ireland and the United States within the group of Anglo-American countries, as shown in panel (f). Interestingly, it also appears that there is more heterogeneity in redistributive preferences across both Western and Eastern Europe than among the group of the Anglo-American countries.

4.3 Distributional Shifts Across Time

Because there are surveys from three different years (i.e. 1987, 1992, and 1999) focusing on questions of social inequality, it is also possible to assess whether individuals' perceptions and normative beliefs have changed over time. However, because only few countries participated in all three surveys, the significance of the corresponding evidence remains somewhat limited. Figure 2 shows density estimates of the demand for equalization of market wages and the two subjective inequality indices, respectively, by year of survey.¹⁴

Figure 2

The first two panels of figure 2 show substantial shifts over time in the distribution of both inequality perceptions and distributional norms. Specifically, inequality perceptions have unambiguously increased over time, as shown in panel (a). This upward shift in inequality perceptions parallels the evolution of the effective level of inequality in most OECD countries (e.g. Atkinson, 2008).¹⁵ At the same time, however, panel (b) shows that the ethical level of inequality has also risen, thus counteracting the upward trend in the perceived level of wage inequality. This is an interesting result, suggesting that the increase in the effective level of inequality was paralleled by a shift towards increasing tolerance to inequality in market

¹⁴For drawing this figure I restrict the sample to observations from those countries/regions that participated in all three surveys: Australia, West Germany, Great Britain, Hungary, and the United States.

 $^{^{15}}$ Also note that there was a bimodal distribution in 1987 that vanished over time. The bimodality is driven by Hungary, the only Eastern European country that participated in the 1987 survey. This in turn implies that subjective evaluations in Hungary must have converged towards the distribution of the other countries over time.

wages. In fact, several authors have argued that such changes in social norms are one important component in understanding the recent rise income inequality (Atkinson, 2003; Levy and Temin, 2007; Piketty and Saez, 2006). Because the distribution of both subjective inequality indices tends to shift upwards over time, they partially cancel each other. As a result, there is actually not much change in the distribution of redistributive preferences over time, as shown in panel (c) of figure 2.

5 Redistributive Preferences and the Effective Level of Inequality and Redistribution

5.1 Baseline Estimates

The next step is to see whether these differences in inequality perceptions, distributional norms, and redistributive preferences across countries and over time are systematically related to corresponding differences in objective measures of redistribution and/or after tax-and-transfer inequality. I begin my examination of this issue by running a series of simple regression models that take the following basic form:

$$\omega_{it} = \beta_0 + \overline{R}_{it}\beta_1 + \epsilon_{it}, \quad \text{or}$$
 (6a)

$$\omega_{jt} = \beta_0 + \overline{G}_{jt}^{\text{actual}} \beta_2 + \overline{G}_{jt}^{\text{ethical}} \beta_3 + \epsilon_{jt}, \tag{6b}$$

with the dependent variable ω_{jt} being either the difference between the Gini coefficient in disposable household income before and after taxes and transfer payments, ΔG_{jt} , total social expenditure as a percentage share of GDP, SocExp_{it}, or simply the Gini coefficient in disposable household income after taxes and transfer payments, G_{jt}^{after} in country j and year t. For the moment, the only regressor in most of the specifications shown is the average demand for redistribution within country/region j and year t, denoted by \overline{R}_{jt} , and thus the parameter of key interest in most cases is given by β_1 (see equation (6a)). In some of the specifications, however, I include the mean values of the two subjective inequality indices as main regressors, as indicated by equation (6b). In this case, parameters β_2 and β_3 are of main interest. The

resulting parameter estimates, for both kind of specification, are given in table 3.

Table 3

The first column shows parameter estimates from a regression of ΔG_{jt} on \overline{R}_{jt} only. The regression yields a statistically significant point estimate of 0.373 (t-value of about 2.56). Note that the estimated coefficient on \overline{R}_{jt} implies a substantive quantitative association between redistributive preferences and the difference in the Gini coefficient. In fact, the approximate elasticity of Δ Gini with respect to redistributive preferences equals about 0.74 (= (0.373 · 0.319)/0.161). The specification in the second column includes country×year means of the two subjective inequality indices, $\overline{G}_{jt}^{\text{actual}}$ and $\overline{G}_{jt}^{\text{ethical}}$, instead of average redistributive preferences. Both coefficients have the expected sign, with a positive coefficient on inequality perceptions and a negative coefficient on distributional norms, and even though they are both insignificant individually, they are jointly significant (see the corresponding F-statistic shown in the bottom row of table 3).

The next two columns show analogous regression specifications for SocExp_{it} as dependent variable, arguably a more direct measure of redistribution than ΔG_{jt} . As above, however, there is a large positive and significant association between redistributive preferences and overall social expenditure. The approximate elasticity is equal to about 0.5 (= $(33.451 \cdot 0.319)/21.253$) in this case, smaller than in the case of ΔG_{jt} , but still surprisingly large. There is also a similar pattern when inequality perceptions and distributional norms are included as regressors, as shown in the fourth column. There is a positive but insignificant effect of inequality perceptions and a negative and significant effect of distributional norms on total social expenditure.

The remaining columns of table 3 show results using the after tax-and-transfer inequality in disposable household income, G_{jt}^{after} , as the dependent variable. In the fifth column, both \overline{R}_{jt} and G_{jt}^{before} are included as regressors. The resulting point estimate on \overline{R}_{jt} in this specification equals -0.352 (with a t-value of about -3.14). Note that, holding inequality before taxes and transfers constant, a negative effect of redistributive preferences on the Gini coefficient after taxes and transfers is consistent with a positive association between redistributive preferences and the difference between inequality before and after taxes/transfers. The sixth column shows, however, that virtually the same estimate results if only \overline{R}_{jt} is included as regressor. The

corresponding estimates suggest that redistributive preferences and the Gini coefficient before taxes/transfers are only weakly correlated with each other. I again include the two subjective inequality indices instead of redistributive preferences in the seventh column. As before, the corresponding point estimates are insignificant individually, but the overall F-statistic shows that they are jointly significant. Even though they are both insignificant, both coefficients have the expected sign (a positive sign for inequality perceptions and a negative sign in the case of distributional norms). Finally, because the Gini_{after} is available for a somewhat larger sample than the Gini_{before}, and because the comparison between the specifications from the fifth and the sixth column shows that the estimated parameter on \overline{R}_{jt} is almost the same, irrespective of whether G_{jt}^{thefore} is (also) included as regressor, it is possible to use a slightly larger sample if Gini_{before} is excluded from the set of regressors (i.e. the sample increases from 35 to 38 observations; see table 1). Not surprisingly, using this extended sample yields a somewhat different point estimate on redistributive preferences ($\hat{\beta}_1 = -0.208$), but one that is not statistically different from that obtained from the smaller sample.

5.2 Subjective Inequality Measures and Individuals' Political Preferences

A closely related question is how individual-level inequality perceptions, distributional norms, and redistributive preferences are linked to the effective level of inequality and redistribution in the short run. There are various potential channels linking subjective and objective measures of inequality and redistribution, but the most likely channel runs from individuals' redistributive preferences to their more general political preferences and their voting behavior (e.g. Borck, 2007). In this part of the analysis, I will thus focus on the empirical association between subjective inequality measures and individuals' more general political preferences such as their support for progressive taxation.

Since I can rely on individual-level data in this case, it is possible to apply a standard twoway fixed-effects regression specification to estimate the strength of the association between individuals' political preferences and their subjective inequality evaluations:

$$p_{ijt} = \beta_0 + x_{ijt}\beta + \omega_{ijt}\alpha + \psi_j + \lambda_t + \varepsilon_{ijt}, \tag{7}$$

where the dependent variable p_{ijt} is one of three distinct measures of individuals' more general political preferences: either individuals' stated support for redistribution by the state, their support for progressive taxation, or a self-assessment of their general political orientation on a simple scale running from far-left to far-right, labeled conservative political orientation below (see appendix A for the exact wording of the underlying survey questions). All specifications either include the two subjective inequality measures or, alternatively, individuals' assessment of market justice, denoted by ω_{ijt} . Moreover, all specifications include a set of individual-level controls, denoted by x_{it} , as well as a full set of country and survey-year fixed effects, denoted by ψ_j and λ_t , respectively.¹⁶

Table 4

Parameter estimates are shown in 4. The first two columns of table 4 shows results for individuals' general support for redistribution by the state. The first column shows that those individuals with a high demand for equalization of market wages tend to be more likely to support the redistribution by the state. The corresponding point estimate is statistically significant and moreover large in substantial terms. Specifically, evaluated at sample means, the elasticity of the support for redistribution by the state with respect to the demand for equalization of market wages amounts to about 0.125 (= $(0.319 \cdot 0.255)/0.65$). The two subjective inequality indices also yield significant point estimates, as shown in the second column. The corresponding elasticities are about 0.258 and -0.256, respectively.

The next two columns show analogous results for individuals' support for progressive taxation. The third column shows a strong and statistically significant association between redistributive preferences and individuals' support for progressive taxation. This shows that

 $^{^{16}}$ The full list of individual-level control variables is as follows: a set of four variables describing an individual's beliefs about the factors that determine and that should determine one's income, an individual's rank within the income distribution within country j and year t, an index of experienced social mobility in the past ten years, an individuals' perception of social conflicts, and a few standard socio-demographic controls such as age or education. See appendix A for additional details.

individuals with a high demand for equalizing market wages are clearly more in favor of progressive taxation than those with a low demand. Again, the size of the estimated coefficients is remarkably large, even though much smaller than in the case of one's general support for redistribution by the state. The elasticity of support for progressive taxation with respect to redistributive preferences amounts to 0.047. Not surprisingly, the two subjective inequality measures are also significantly associated with the support for progressive taxation, as shown in the fourth column.

Finally, the last two columns report results for individuals' political self-assessment on a simple left-right scale, where higher values on the scale denote a more conservative orientation. As expected, individuals with a higher demand for equalizing market wages are less likely to think of themselves as having a conservative orientation. Again, besides being statistically significant, the estimated coefficient turns out to be large in substantive terms. The elasticity of conservative orientation with respect to the demand for equalization of market wages is -0.156, while the elasticities with respect to the mean of actual and ethical inequality equal -0.328 and 0.323, respectively.

5.3 Controlling for Additional Country Characteristics

Taken together, the empirical evidence presented so far is clearly consistent with the view that we observe different equilibria of redistributive preferences and the effective level of redistribution across countries and time, with a high (low) demand for redistribution going hand in hand with a high (low) effective level of redistribution and a low (high) level of after tax-and-transfer inequality. There is also evidence that redistributive preferences (alternatively, inequality perceptions and distributional norms) are linked to the effective, aggregate level of inequality and redistribution though individuals' more general political preferences and their voting behavior.

In the medium and longer run, however, one also needs to explain why countries end up in one equilibrium and not the other. This will immediately shift our focus to more fundamental, and often non-economic, factors such as a country's political institutions. For example, in a comprehensive review of the theoretical literature and the empirical evidence explaining the different social security systems of the US and Europe, Alesina and Glaeser (2004, p.3)

conclude that "(...) economic considerations alone do not go very far in explaining American exceptionalism". More specifically, they argue that political institutions and ethnic diversity, rather than purely economic factors, are the key to ultimately understand the difference between the US and Europe in terms of after tax-and-transfer inequality and redistribution. I will thus next examine if the simple association between redistributive preferences and the effective level of inequality/redistribution still holds if arguably more fundamental country characteristics are taken into account.

In the following, given the somewhat limited degrees of freedom available for the aggregatelevel analysis, I restrict myself to a few variables that have been prominently discussed in the existing literature and mainly follow the empirical analysis done by Alesina et al. (2001) and Alesina and Glaeser (2004). One factor that has received a lot of attention among economists is the before tax-and-transfer inequality in income, hypothesized to have a positive effect on the extent of redistribution (e.g. Borge and Rattsø, 2004; Milanovic, 2000). The age distribution in the population has also been suggested as an important explanatory factor, since many redistributive policies are mechanically linked to the age distribution in the population; pensions are the most prominent example of this (Galasso and Profeta, 2007; Razin et al., 2002; Tabellini, 2000). Another well-known hypothesis is that countries that are more open to trade redistribute more because they are exposed to more aggregate income volatility. It is hypothesized that these countries therefore use redistributive policies to insure individuals against the increased risk of negative income shocks (e.g. Rodrik, 1998). Moreover, several researchers have argued that additional, non-economic factors are even more important for our understanding of country differences in after tax-and-transfer inequality and redistribution. Among the factors identified as most important are features of a country's political institutions and especially the existence of proportional representation (e.g. Alesina and Glaeser, 2004; Alesina et al., 2001). It has also been shown that ethnic fractionalization of the electorate has a strong negative impact on the extent of redistribution (Alesina et al., 1999; Luttmer, 2001). Finally, the historical experience

¹⁷The intuition is based on the median-voter hypothesis (e.g. Meltzer and Richard, 1981), which basically states that the demand for redistribution is greater, the larger the inequality before taxes and transfers. Moreover, note that treating the before taxes-and-transfer inequality may be elusive because, for example, labor market regulations and institutions obviously have an impact on the before tax-and-transfer income distribution and are likely to be associated with individuals' redistributive preferences.

of having been exposed to socialist ideology and rule may have had a long-lasting impact on both (the functioning of) institutions and as well as on individuals' trust in these institutions (e.g. Alesina and Fuchs-Schündeln, 2007).

Based on these considerations I run a simple variation of the baseline regression specification from before:

$$\omega_{it} = \beta_0 + \overline{R}_{it}\beta_1 + x_{it}\beta + \lambda_t + \epsilon_{it}, \tag{8}$$

which corresponds exactly to equation (6a), with the exception that I now take additional country characteristics x_{jt} as well as survey-year fixed effects λ_t into account. Moreover, for the sake of brevity, I focus on redistributive preferences as main regressor in this part of the analysis. More specifically, based on the arguments above, I include the following set of country-level controls in all specifications:¹⁸ the population share aged 65 or over, openness to trade, a binary indicator for proportional representation, an index of ethnic fractionalization, and a binary indicator for countries that were formerly under socialist rule. The Gini coefficient of disposable household income before taxes and transfers is only included in one of the specifications for the Gini coefficient after taxes and transfers.¹⁹

Table 5

The resulting parameter estimates are given in table 5. The dependent variable in the first specification is again the difference between the Gini coefficient before and after taxes and transfer payments, Δ Gini. Controlling for additional country characteristics, the coefficient on \overline{R}_{jt} turns out to be much smaller than in the baseline specification (see table 3). The resulting point estimate equals -0.049, which is statistically insignificant (t-value of -0.271). In contrast, however, there is a significant negative coefficient of ethnic fractionalization and a significant positive coefficient of the dummy variable for former socialist countries. The other coefficients

¹⁸See appendix table A.1 for details and data sources.

 $^{^{19}}$ In the case of Δ Gini, there is a correlation with Gini_{before} simply by construction. Moreover, one may argue that the Gini coefficient before taxes/transfers is endogenous and should rather be viewed as an additional outcome (cf. footnote 17). Similarly, average income in a country is arguably endogenous, and I thus only discuss estimation results without controlling for per-capita income, but I get very similar results if I also control for a country's average income (i.e. log per-capita GDP).

turn out to be statistically insignificant as well. Moreover, note that the model fit is nonetheless good in the sense that the model has good predictive power overall (R-squared of 0.57 with six regressors only).

The second column replicates the specification from the first column, but uses total social expenditure as dependent variable. Similarly, the coefficient on redistributive preferences becomes insignificant once additional country characteristics are taken into account. The point estimate is only about a quarter of the baseline estimate from table 3 above, with a small t-value of 0.522. Again, most other regressors turn out be insignificant as well, with the exception of the percentage share of the country' population that is aged 65 or above in this case.

The remaining three columns show results for the Gini in disposable household income after taxes and transfer payments. The third column shows a small and insignificant estimate on redistributive preferences ($\hat{\beta}_1 = -0.061$, with a t-value of -0.356) as well as on the Gini before taxes and transfers when additional controls are included. The fourth column again shows that virtually the same point estimates result if the Gini coefficient in disposable household income before taxes and transfers is omitted from the list of controls (point estimate of -0.101, t-value of -0.652). Finally, the fifth column shows results for the same specification but for the slightly larger sample (as in the baseline estimates). Parameter estimates again turn out to be very similar. The coefficient on redistributive preferences remains small and statistically insignificant (the corresponding point estimate equals 0.054, with a t-value of about 0.5).

6 Conclusions

This paper shows that there is considerable variation in individuals' inequality perceptions, distributional norms, and redistributive preferences, both within and across countries/regions. Not surprisingly, individuals from European countries tend to demand more redistribution than those from Anglo-American countries. However, the descriptive analysis also shows that there is considerable heterogeneity across different countries from within Eastern or Western Europe or among the group of Anglo-American countries. There is also evidence of substantial upward shifts in both inequality perceptions and distributional norms over time (i.e. a shift towards more tolerance vis-à-vis inequality in market wages).

However, the main focus of the empirical analysis is on the empirical association between the two subjective inequality measures and redistributive preferences on the one hand, and measures of the effective level of inequality and redistribution on the other hand. I find that both differences in subjective inequality indices and redistributive preferences are strongly and significantly associated with objective measures of inequality and redistribution, in line with the hypothesized equilibrium between redistributive preferences the effective supply of redistribution of the theoretical models outlined in the introduction (e.g. Alesina and Angeletos, 2005; Bénabou and Tirole, 2006). There is also evidence that the subjective and objective dimension of inequality and redistribution are, at least in part, linked through individuals' more general political preferences. Indeed, and in line with the evidence of strong associations between subjective and objective inequality measures, individuals with stronger redistributive preferences tend to be more likely to be in favor of state intervention in order to reduce the level of inequality, more likely to be supportive of progressive taxation, as well as less likely to state a conservative general political orientation.

I finally show that the simple association between redistributive preferences and the effective level of redistribution or after tax-and-transfer inequality vanishes if additional, and arguably more fundamental, country characteristics are taken into account. These additional results suggest that more fundamental and more deeply rooted country characteristics such as a country's political institutions explain both country differences in redistributive preferences (or, alternatively, differences in inequality perceptions and distributional norms) as well as country differences in the effective level of after-tax/transfer inequality and the effective extent of redistribution.

References

- Alesina, A. and Angeletos, G.-M. (2005). Fairness and Redistribution. *American Economic Review*, **95**(4), 960–980.
- Alesina, A. and Fuchs-Schündeln, N. (2007). Good-bye Lenin (or not?): The effect of communism on people's preferences. *American Economic Review*, **97**(4), 1507–1528.
- Alesina, A. and Giuliano, P. (2009). Preferences for Redistribution. NBER Working Paper No. 14825.
- Alesina, A. and Glaeser, E. (2004). Fighting Poverty in the US and Europe: A World of Difference. Oxford University Press.
- Alesina, A. and La Ferrara, E. (2005). Preferences for Redistribution in the Land of Opportunities. *Journal of Public Economics*, **89**(5-6), 897–931.
- Alesina, A., Baqir, R., and Easterly, W. (1999). Public Goods and Ethnic Divisions. *Quarterly Journal of Economics*, **114**(4), 1243–1284.
- Alesina, A., Glaeser, E., and Sacerdote, B. (2001). Why doesn't the united states have a european-style welfare state? *Brookings Papers on Economic Activity*, **2001**(2), 187–254.
- Atkinson, A. (2003). The changing distribution of income: evidence and explanations. German Economic Review, 1(1), 3–18.
- Atkinson, A. (2008). The Changing Distribution of Earnings in OECD Countries. Oxford University Press.
- Bénabou, R. and Tirole, J. (2006). Belief in a Just World and Redistributive Politics. *Quarterly Journal of Economics*, **121**(2), 699–746.
- Borck, R. (2007). Voting, Inequality and Redistribution. *Journal of Economic Surveys*, **21**(1), 90–109.
- Borge, L. and Rattsø, J. (2004). Income distribution and tax structure: Empirical test of the Meltzer-Richard hypothesis. *European Economic Review*, **48**(4), 805–826.
- Corneo, G. and Grüner, H. P. (2002). Individual Preferences for Political Redistribution. Journal of Public Economics, 83, 83–107.
- Doerrenberg, P. and Peichl, A. (2012). The impact of redistributive policies on inequality in oecd countries. IZA Discussion Paper No. 6505.
- Fong, C. (2001). Social Preferences, Self-Interest and the Demand for Redistribution. *Journal of Public Economics*, **82**, 225–246.
- Galasso, V. and Profeta, P. (2007). How does ageing affect the welfare state? European Journal of Political Economy, 23(2), 554–563.
- Gastwirth, J. and Glauberman, M. (1976). The Interpolation of the Lorenz Curve and Gini Index from Grouped Data. *Econometrica*, **44**(3), 479–483.

- Isaksson, A. and Lindskog, A. (2009). Preferences for redistribution—A country comparison of fairness judgements. *Journal of Economic Behavior & Organization*, **72**(3), 884–902.
- Jasso, G. (1980). A New Theory of Distributive Justice. American Sociological Review, 45(1), 3–32.
- Kuhn, A. (2011). In the eye of the beholder: Subjective inequality measures and individuals' assessment of market justice. *European Journal of Political Economy*, **27**(4), 625–641.
- Levy, F. and Temin, P. (2007). Inequality and Institutions in 20th Century America. NBER Working Paper No. 13106.
- Luttmer, E. (2001). Group loyalty and the taste for redistribution. *Journal of Political Economy*, **109**(3), 500–528.
- Meltzer, A. H. and Richard, S. F. (1981). A Rational Theory of the Size of Government. *Journal of Political Economy*, **89**(5), 914–27.
- Milanovic, B. (2000). The median-voter hypothesis, income inequality, and income redistribution: an empirical test with the required data. *European Journal of Political Economy*, **16**(3), 367–410.
- OECD (2008). Growing Unequal? Income Distribution and Poverty in OECD Countries. Technical report, OECD.
- Osberg, L. and Smeeding, T. (2006). "Fair" Inequality? Attitudes toward Pay Differentials: The United States in Comparative Perspective. American Sociological Review, **71**(3), 450–473.
- Piketty, T. (1995). Social mobility and redistributive politics. *Quarterly Journal of Economics*, **110**(3), 551–584.
- Piketty, T. and Saez, E. (2006). The evolution of top incomes: a historical and international perspective. *American Economic Review*, **96**(2), 200–205.
- Ravallion, M. and Lokshin, M. (2000). Who wants to redistribute? The tunnel effect in 1990s Russia. *Journal of Public Economics*, **76**(1), 87–104.
- Razin, A., Sadka, E., and Swagel, P. (2002). The aging population and the size of the welfare state. *Journal of Political Economy*, **110**(4), 900–918.
- Rodrik, D. (1998). Why do more open economies have bigger governments? *Journal of Political Economy*, **106**(5), 997–1032.
- Tabellini, G. (2000). A positive theory of social security. Scandinavian Journal of Economics, 102(3), 523–545.

Table 1: Individual and aggregate number of valid observations, by country/region and by year of survey

Country	1987	28	19	1992	16	1999	${ m To}$	Total
Australia	1,387	(1,1)	1,956	(1,1)	1,477	(1,1)	4,820	(3,3)
Austria	891	(1,1)			872	(1,1)	1,763	(2,2)
Canada			871	(1,1)	968	(1,1)	1,767	(2,2)
Czech Republic			298	(1,1)	1,665	(1,1)	2,532	(2,2)
France					1,701	(1,1)	1,701	(1,1)
East Germany			861	(1,1)	432	(1,1)	1,293	(2,2)
West Germany	1,153	(1,1)	1,477	(1,1)	737	(1,1)	3,367	(3,3)
Great Britain	1,009	(1,1)	912	(1,1)	621	(1,1)	2,542	(3,3)
Hungary	2,083	(0,0)	1,096	(0,1)	1,000	(0,1)	4,179	(0,2)
Italy			940	(1,1)			940	(1,1)
Netherlands	1,405	(1,1)					1,405	(1,1)
New Zealand			1,091	(1,1)	962	(1,1)	2,056	(2,2)
Northern Ireland					514	(1,1)	514	(1,1)
Norway			286	(1,1)	1,153	(1,1)	2,140	(2,2)
Poland			1,112	(1,1)	895	(1,1)	2,007	(2,2)
Portugal					745	(1,1)	745	(1,1)
Slovakia					1,059	(1,1)	1,059	(1,1)
Spain					720	(0,1)	720	(0,1)
Sweden			574	(1,1)	983	(1,1)	1,557	(2,2)
Switzerland	829	(1,1)					829	(1,1)
United States	1,184	(1,1)	843	(1,1)	991	(1,1)	3,018	(3,3)
Total	9,941	(7,7)	13,587	(12,13)	17,426	(16,18)	40,954	(35,38)

equality indices. Valid observations at the aggregate level are those with non-missing values on both before and after tax-and transfer inequality and on after tax-and-transfer inequality only, observations at the individual-level are those with non-missing values on the two subjective in-Notes: The table shows individual and aggregate (in parentheses) number of observations. Valid respectively (first and second number in parentheses, respectively).

Table 2: Subjective inequality indices and redistributive preferences, descriptive statistics

	Mean	Standard deviation
(a) Moments of subjective wage distributions		
t optimation shares Population share bottom group	0.774	0.089
r optimion simio, bosom group	# · · ·	100:0
Population share, top group	0.226	0.082
Wage Sitales	0	1177
Actual wage share, bottom group	0.430	0.174
Actual wage share, top group	0.570	0.174
Ethical wage share, bottom group	0.544	0.166
Ethical wage share, top group	0.456	0.166
(b) Subjective inequality indices		
Inequality perceptions	0.451	0.192
$1(\operatorname{Actual}\operatorname{Gini}<0)$	0.001	0.025
$1(Actual\ Gini = 0)$	0.000	0.000
$1(\operatorname{Actual} \operatorname{Gini} > 0)$	0.999	0.025
Distributional norms	0.303	0.175
1 (Ethical Gini < 0)	0.004	0.060
1 (Ethical Gini = 0)	0.006	0.076
1 (Ethical Gini > 0)	0.991	0.096
(c) Redistributive preferences		
Demand for equalization of market wages	0.322	0.269
$1(\mathrm{Demand} < 0)$	0.072	0.259
$1(\mathrm{Demand}=0)$	0.017	0.129
$1(\mathrm{Demand} > 0)$	0.911	0.285
$1(\mathrm{Demand}=1)$	0.006	0.076
1(Demand > 1)	0.003	0.056

Notes: All numbers refer to the data pooled across all countries/regions and years (n = 40,954). 1(·) denotes the indicator function, taking the expression in parentheses as argument. Variable definitions are given in the main text (see section 3).

Table 3: Subjective inequality indices, redistributive preferences, inequality, and redistribution

	$\Delta \mathrm{Gini}$	ii	SocExp	dx5		Gini _{after}	ter	
Mean Standard deviation	0.161	0.161	21.253	21.253 4.748	0.296	0.296	0.296	0.297
Redistributive preferences	0.373^{**} (0.146)		33.452^{*} (18.048)		$-0.352^{\star\star\star}$ (0.112)	-0.344^{***} (0.111)		-0.208^{\star} (0.108)
${ m Gini}_{ m before}$					0.263^{*} (0.146)			
Inequality perceptions		0.299 (0.357)		44.811 (31.991)			-0.035 (0.248)	
Distributional norms		-0.619 (0.383)		-71.315^{\star} (39.267)			0.326 (0.301)	
Constant	0.042 (0.047)	0.218^{***} (0.057)	10.596^{*} (5.687)	$23.051^{\star\star\star}$ (4.119)	$0.288^{\star\star\star}$ (0.079)	0.406^{***} (0.035)	$0.210^{\star\star\star}$ (0.026)	0.365^{***} (0.035)
Number of observations p-value (F-statistic) R-Squared	35 0.015 0.155	35 0.012 0.234	35 0.073 0.166	35 0.057 0.200	35 0.002 0.254	35 0.004 0.192	35 0.000 0.355	38 0.062 0.097

Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Robust standard errors are given in parentheses. Variable definitions are given in the main text (section 3) and appendix table A.1.

Table 4: Subjective inequality indices, redistributive preferences, and political preferences

	Support for redistribution by the state	t for by the state	Support for progressive taxation	rt for taxation	Conservative political orientation	ntive entation
Mean Standard deviation	0.625	0.625	0.770	0.770	0.339	0.339
Redistributive preferences	0.289***		0.121***		-0.192^{***} (0.012)	
Inequality perceptions		0.417^{***} (0.017)		0.266^{***} (0.016)	,	-0.269^{***} (0.023)
Distributional norms		-0.633^{***} (0.019)		-0.272^{***} (0.018)		0.352^{***} (0.024)
Individual-level controls Country fixed effects Survey-year fixed effects	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes	Yes Yes Yes
Number of observations p-value (F-statistic) R-Squared	39,803 0.000 0.191	39,803 0.000 0.190	38,582 0.000 0.056	38,582 0.000 0.058	$24,244 \\ 0.000 \\ 0.103$	24,244 0.000 0.102

Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Robust standard errors are given in parentheses. See appendix A for the full list of and additional information on individual-level controls.

Table 5: Controlling for additional country characteristics

	$\Delta \mathrm{Gini}$	SocExp		${ m Gini}_{ m after}$	
Mean Standard deviation	0.161	21.253	0.296	0.296	0.297
Redistributive preferences	-0.026	7.690	-0.050 (0.157)	-0.061	0.099
${ m Gini}_{ m before}$	(651.6)		0.123 0.256)	(111:0)	
Population share 65+	-0.005	$1.374^{\star\star\star}$	-0.005	_0.006**	-0.005^{*}
	(0.005)	(0.362)	(0.004)	(0.003)	(0.003)
Openness	0.021	1.548 (3 633)	-0.049^{***}	-0.053^{**}	-0.065^{***}
Proportional representation	0.025	(2.725)	(0.01) -0.050**	-0.054^{***}	-0.048^{**}
	(0.020)	(1.731)	(0.021)	(0.019)	(0.018)
Ethnic fractionalization	$-0.128^{\star\star}$	-1.730	-0.033	-0.056	-0.020
	(0.049)	(4.683)	(0.061)	(0.038)	(0.039)
East	0.049^{**}	-0.714	-0.013	-0.008	-0.010
	(0.019)	(1.814)	(0.021)	(0.015)	(0.013)
Constant	$0.212^{\star\star\star}$	-3.454	0.393^{*}	$0.478^{\star\star\star}$	0.412^{***}
	(0.073)	(6.668)	(0.191)	(0.051)	(0.050)
Survey-year fixed effects	Yes	Yes	Yes	Yes	Yes
Number of observations	35	35	35	35	38
p-value (F-statistic)	0.000	0.000	0.000	0.000	0.000
R-Squared	0.583	0.628	0.591	0.587	0.527

Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Robust standard errors are given in parentheses. Variable definitions are given in the main text (section 3) and appendix table A.1.

--- Western Europe (c) Redistributive preferences ----- Anglo-american countries Eastern Europe ٦.5 Ğ. S 3.1 д. Density 1 Density 1 --- Western Europe (b) Distributional norms ------ Anglo-american countries Eastern Europe Density 2 2.1 0 Density 1 3.5 €. ε 2.5 Š 3.1 ζ. 0 --- Western Europe (a) Inequality perceptions ----- Anglo-american countries Eastern Europe Vaisne Density 2.5 3.1 Valensity 1 ς.

Figure 1: Subjective inequality measures, by region or by country within region

Notes: Panels (a) to (c) show density estimates by broad region, while panels (d) to (f) show density estimates by country (or by region within country); see also table 1. Variable definitions are given in the main text (see section 3). All density estimates use the Gaussian kernel.

(e) Redistributive preferences, Eastern Europe

(d) Redistributive preferences, Western Europe

---- Switzerland

---- Sweden

(f) Redistributive preferences, Anglo-American coun-

-- United States

---- Northern Ireland

---- Slovakia

-- East Germany

1999 (c) Redistributive preferences **———** 1992 **—** 1987 0 ð.1 1.25 Ğ. 25. 1999 (b) Distributional norms **———** 1992 1987 Valiene Density 5.1 2.5 ς. 0 Š 1999 (a) Inequality perceptions **———** 1992 Actual inequality 1987 Density 5.1 2.5

Figure 2: Subjective inequality measures, by year of survey (selected set of countries only)

Notes: Figures are based on observations from those countries that participated in all three years only (Australia, former West Germany, Great Britain, Hungary, and the United States). Variable definitions are given in the main text (see section 3). All density estimates use the Gaussian kernel.

A Variable Definitions and Data Sources

A.1 Aggregate-level Variables

Inequality and redistribution Mainly due to reasons of data availability, I focus on the Gini coefficients in disposable household income before and after taxes and transfer payments. I further use either the difference between the two Gini coefficients or overall social expenditure (as a percentage share of GDP) as my measure of redistribution.

Aggregate-level controls In the aggregate-level regressions (see section 5.3), I include the percentage share of the population aged 65 or above, a country's openness, an indicator if a country has proportional representation, an index of ethnic fractionalization, and an indicator if a country was under socialist rule after World War II. See table A.1 for details.

A.2 Political Preferences

Support for redistribution through the state: Individuals' support for redistribution by the state is measured by a simple item from the survey: "Do you agree or disagree? It is the responsibility of the government to reduce the differences in income between people with high incomes and those with low incomes." The possible answers range from 1 ("strongly disagree") to 5 ("strongly agree"). In the empirical analysis, I simply use a dichotomized variable taking on the value 1 if an individual (strongly) agrees with the statement and 0 otherwise.

Support for progressive taxation: Individuals' support for progressive taxation is measured by a single survey item: "Do you think people with high incomes should pay a larger share of their income in taxes than those with low incomes, the same share, or a smaller share?" Possible answers range from 1 ("much smaller share") to 5 ("much larger share"). In the analysis, I again use a dichotomized variable taking on the value 1 if an individual states that people with high incomes should pay a (much) larger share, and 0 otherwise.

Conservative orientation: Individuals' stated political orientation on a simple left-right scale. "Based on your current voting intention, would you consider yourself left-wing or right-wing?" Possible answer categories range from 1 ("far left") to 5 ("far right"). In the analysis I use a dichotomized variable taking on the value 1 if an individual calls him- or herself (far) right.

A.3 Individual-level Regressors

I use three sets of regressors to model the variation in individuals' more general political preferences (section 5.2): (i) variables which describe the factors that do or ought to determine actual pay from the individuals' point of view, (ii) variables describing individuals' self-interest in redistribution or economic variables mediating self-interest in redistribution, and (iii) additional control variables which potentially correlate with unobserved determinants of subjective inequality measures (like risk aversion, for example).

Belief that needs should be important: This variable is meant to capture the extent to which an individual thinks that one's needs should be important in determining his or her income (need principle). This variable is constructed from the following two questions about

the factors that should be important in determining one's pay: "In deciding how much people ought to earn, how important should each of these things be, in your opinion? (i) What is needed to support your family. (ii) Whether the person has children to support."

Belief that effort should be important: This variable is meant to capture the equity principle and is constructed from five questions about which factors should be important in determining pay: "In deciding how much people ought to earn, how important should each of these things be, in your opinion? (i) How much responsibility goes with the job. (ii) The number of years spent in education and training. (iii) Whether the job requires supervising others. (iv) How well he or she does the job. (v) How hard he or she works at the job."

Perception that ascribed skills are important: This variable measures the extent to which a person beliefs in ascribed factors as being important in determining the amount of compensation. This question relates to the perception of individuals of which factors actually are important for getting ahead. "We have some questions about opportunities for getting ahead: (i) How important is coming from a wealthy family? (ii) Knowing the right people?"

Perception that acquired skills are important: This variable measures if an individual thinks that acquired skills are actually important in determining one's pay. The variable is the sum of over two different questions: "We have some questions about opportunities for getting ahead: (i) Do you agree or disagree? In [country], people get rewarded for their effort. (ii) In [country], people get rewarded for their intelligence and skills."

Income: An individual's personal net monthly income, expressed as the rank of an individual within the income distribution in a given country and year (to remove different scaling).

Social mobility: There are two simple questions about individuals' self-perception of the position today and ten years ago: "(i) In our society, there are groups which tend to be towards the top and groups which tend to be toward the bottom. Below is a scale that runs from top to bottom. Where would you put yourself on this scale? (ii) And ten years ago, where did you fit then?" Both are measured on a scale from 1 (bottom) to 10 (top). The mobility index is simply defined as the difference between the two scores (i.e. position today minus position ten years ago).

Perception of social conflicts: This variable measures individuals' perceptions of conflicts within society. Included items are questions about the existence of conflicts between: "In all countries, there are differences or even conflicts between different social groups. In your opinion, In [country] how much conflict is there between...: (i) Poor people and rich people? (ii) The working class and the middle class? (iii) Management and workers? (iv) People at the top of society and people at the bottom? (v) Young people and older people?"

Socio-demographic controls: The remaining controls that are used in the analysis are self-explanatory: Age (in years), highest attained education (in years), gender, occupation (dummy variables denoting major occupational group according to the International Standard Classification of Occupations), and employment status (employment, unemployment, and nonemployment).

Table A.1: Definition, basic descriptives and data source for aggregate-level variables

Variable label	Definition	Mean	Standard deviation Source	Source
Gini _{before}	Gini of disposable household income, before taxes and transfers	0.457	0.043	World Income Inequality Database
$\operatorname{Gini}_{\operatorname{after}}$	Gini of disposable household income, after taxes and transfers	0.296	0.045	World Income Inequality Database
$\Delta Gini$	Gini _{before} - Gini _{after}	0.161	0.055	World Income Inequality Database
SocExp	Overall social expenditure (% of GDP)	21.253	4.748	World Bank
Popshare 65+	Percentage of population aged 65 or older	13.991	2.058	World Bank
Openness	Export share (exports as share of GDP)	0.595	0.249	World Bank
	+ import share (imports as share of GDP)			
Proportional representation	Indicator variable $(= 1 \text{ if country has})$	0.086	0.471	Database of Political Institutions
	proportional representation)			
Ethnic fractionalization	Probability that two randomly chosen	0.222	0.188	World Bank
	individuals have different ethnicity			
East	Indicator variable $(= 1 \text{ if country has been})$	0.200	0.406	1
	under socialist rule after World War II)			

Notes: All descriptives refer to the sample of observations for which the Gini in disposable household income both before and after taxes and transfer payments is observed (n = 35, see also table 1).

B Estimating the Components of Subjective Gini Coefficients

B.1 Subjective Wage Estimates

The most fascinating data available in the ISSP surveys focusing on social inequality are individuals' subjective estimates of actual and ethical wages for people working in various specific occupations such as a bus driver, a skilled worker in a factory, or a doctor in general practice. More specifically, individuals were asked to estimate what they thought to be the actual wage of people working in different occupations and what they thought people in these occupations ought to be paid from their point of view, denoted as actual and ethical wage estimates for short in the following (original wording from the source questionnaire):

"We would like to know what you think people in these jobs actually earn. Please write how much you think they actually earn each month (before taxes, but after social security contributions). Many people are not exactly sure about this, but your best guess will be close enough."

"Next, what do you think people in these jobs ought to be paid. How much do you think they should earn each month (before taxes, but after social security contributions), regardless of what they actually earn."

Table B.1 contains the full list of occupations for which subjective wage estimates are available (along with the original description of the various occupations in the survey). In the following I briefly describe how I aggregate these wage estimates so that they fit equation (3a) and (3b), respectively. Additional details concerning the conceptual framework are available from Kuhn (2011).

B.2 Group-Specific Wages

I first assign each occupation for which individual wage estimates are available to either the bottom or the top group of wage earners:

bottom = {unskilled worker, farm worker, shop assistant, secretary, bank clerk, bus driver, skilled worker, bricklayer, owner of a small shop} (B.1a)

$$top = \{doctor, lawyer, minister, judge, chairman, owner of a factory\}$$
(B.1b)

Based on this simple classification, group-specific wage estimates (actual and ethical) are constructed as simple within-group averages of occupation-specific wage estimates:

$$\overline{y}(i)_{\text{bottom}}^{w} = \frac{\sum_{j=1}^{15} \mathbf{1}(j \in \text{bottom}) y(i)_{j}^{w}}{\sum_{j=1}^{15} \mathbf{1}(j \in \text{bottom}) \mathbf{1}(y(i)_{j}^{w} \neq *)}, \quad \text{and}$$
(B.2a)

$$\overline{y}(i)_{\text{top}}^{w} = \frac{\sum_{j=1}^{15} \mathbf{1}(j \in \text{top}) y(i)_{j}^{w}}{\sum_{j=1}^{15} \mathbf{1}(j \in \text{top}) \mathbf{1}(y(i)_{j}^{w} \neq *)},$$
(B.2b)

with $y(i)_j^w$ denoting an individual's actual or ethical wage estimate for people working in occupation j, with $w \in \{\text{actual}, \text{ethical}\}$, and with * denoting a missing wage estimate. As indicated by the second indicator function in the denominator, group-specific wages can be computed for any single individual as long as he or she gives at least one wage estimate per group (i.e. at

least one wage estimate for the top group and at least one estimate for the bottom group). This procedure can also easily take care of the fact that the list of occupations has changed over time and that wage estimates for all fifteen occupations are available only in 1992 and only for a few countries (see appendix table B.1).

B.3 Fraction of Individuals Belonging to the Bottom Group

Finally, the fraction of individuals belonging to the bottom group is estimated from the distribution of individuals across occupations in the sample:

$$f_{\text{bottom}} = \frac{1}{n} \sum_{i=1}^{n} \mathbf{1}(\text{isco}_i \in [3, 9]),$$
 (B.3)

with isco_i denoting an individual's major occupational code according to the International Standard Classification of Occupations (ISCO). According to this classification, major group 1 consists of "legislators, senior officials and managers" and major group 2 of "professionals". Note that I estimate a different f_{bottom} for each pair of country and year. As mentioned above, however, within any pair of country×year the size of the two groups is the same for all individuals and the same for both the distribution of actual and ethical wages. In fact, however, estimated group shares do not change much over time and thus results would hardly change if I would allow the population shares to vary only across countries only (but not over time).

Table B.1: Subjective Wage Estimates for People Working in Specific Occupations

Occupation	Abbreviation	1987	1987 1992 1999	1999
"An unskilled worker in a factory"	Unskilled worker	>	>	>
"A bricklayer"	Bricklayer	>	2	
"A shop assistant in a department store"	Shop assistant		>	>
"A farm worker"	Farm worker	>	>	
"A secretary"	Secretary	>	2	
"A bus driver"	Bus driver	>	5	
"A skilled worker in a factory"	Skilled worker	>	>	>
"A bank clerk"	Bank clerk	>	2	
"The owner of a small shop"	Shop owner	>	>	
"A lawyer"	Lawyer		>	>
"A doctor in general practice"	Doctor	>	>	>
"The owner-manager of a large factory"	Owner of a factory		>	>
"A cabinet minister in the national government"	Minister	>	>	>
"A judge in [country's highest court]"	Judge		>	>
"The chairman of a large national company"	Chairman	>	>	>
"Your own occupation"	Own occupation			>

Notes: The first column contains the original wording from the ISSP source questionnaire. The second column lists the corresponding abbreviations used in the figures and tables. (\checkmark) denotes that wage estimates in the 1992 survey are available for a couple of countries (i.e. Australia, Sweden, and Slovenia) only.